ASSESSMENT OF PUBLIC PROCUREMENT POLICY IMPLEMENTATION IN SELECTED SECONDARY SCHOOLS IN KASUNGU DISTRICT

M. Ed. (POLICY, PLANNING AND LEADERSHIP) THESIS

BEATRICE NYONGOLO

UNIVERSITY OF MALAWI

CHANCELLOR COLLEGE

JUNE, 2018



ASSESSMENT OF PUBLIC PROCUREMENT POLICY IMPLEMENTATION IN SELECTED SECONDARY SCHOOLS IN KASUNGU DISTRICT M. Ed. (POLICY, PLANNING AND LEADERSHIP) THES

By Beatrice Nyongolo

B.Sc. (Education) - Mzuzu University

Submitted to Department of Education Foundations, School of Education, in partial fulfilment of the requirement of the degree of Master of Education (Policy, Planning and Leadership

UNIVERSITY OF MALAWI

CHANCELLOR COLLEGE

JUNE, 2018

DECLARATION

I, the undersigned, hereby declare that this thesis is my own original work which has not been submitted to any other institution for similar purpose(s). Where people's work has been used, acknowledgements have been made.

Full Legal Name
Signature
Date

CERTIFICATE OF APPROVAL

The undersigned certify that this thesis represents the student's own work and effort and has been submitted with our approval.

Signature:	_Date:
Richard Walibwe Nyirongo, PhD (Lectur	er)
Main Supervisor	
Signature:	_Date:
Jubilee Mutiopanji Tizifa, PhD (Lecturer)
Co-supervisor	

DEDICATION

To my son, Bright, who was born during the writing of this thesis, for he was a blessing to our family.

ACKNOWLEDGEMENTS

First and foremost I express my gratitude to the Almighty God for giving me the strength and courage to embark on this research. I am also very grateful to my supervisors, Dr. R. W. Nyirongo and Dr. J.M. Tizifa for their meticulous supervision coupled with their constructive criticism to bring this work into shape. Dr. J. Tizifa has not only been a supervisor but also a mother to me. Her encouragement throughout the thesis writing and her hardworking spirit will forever influence my life. I would also like to express my deepest appreciation to Dr. A. Chauma for his kindness to edit my thesis. I wish to thank Mrs M.Wasili for her great hospitality and encouragement during my studies. I also wish to thank the Education Division Manager for Central East Education Division for allowing me to collect data in secondary schools in Kasungu District. Many thanks go to all head teachers and Internal Procurement Committee (IPC) members from all the schools that I collected data for the unconditional assistance which they accorded me.

Last but not least, I would like to express my sincere gratitude to my husband, Patrick, for his patience and spiritual as well as financial support during my studies. Many thanks to my lovely daughter Blessings, and my son Bright, for their perseverance when I denied them the company and attention they deserved during my studies. To all those who in one way or the other supported me to complete this study, I say thank you. May God richly bless you all!

ABSTRACT

The need for effective implementation of the public procurement policy in public schools is crucial for the success of schools in financial management. The Office of Director of Public Procurement in Malawi has been making efforts to improve the procurement process in education institutions but the process still displays different bases of order and compliance. This study assessed the implementation of the public procurement policy in selected schools in Kasungu District. The study was based on the institutional theory as expressed by Scott (2004) who posits that institutions are composed of regulative, normative and cultural cognitive elements that, together with associated activities and resources give stability and meaning to social life. This was a qualitative study that employed a case study design. Purposive sampling was used to select the schools and the respondents. The study targeted four public secondary schools and the respondents were 24. The respondents were head teachers, deputy head teachers, accounts clerks or teacher bursars and IPC members. Data was generated using semi-structured interviews, focus group discussions and document analysis methods. The findings of the study indicated that the public procurement policy was partly being implemented in public schools. It established that IPC members were failing to fully implement this policy due to lack of registered suppliers, lack of finances, and lack of training and cumbersome procedures in procurement. In addition, the measures that were being used to tackle the challenges created loopholes for corruption in public schools. The implication is that if the IPC members still procure goods and services using the little knowledge that they possess then there will be major losses of public funds in schools.

TABLE OF CONTENTS

ABSTRACT	v
TABLE OF CONTENTS	vi
LIST OF APPENDICES	xi
LIST OF ABBREVIATIONS AND ACRONYMS	xii
CHAPTER 1	1
INTRODUCTION	1
1.0 Chapter overview	1
1.1 Background to the study	1
1.1.1 The Public procurement policy	2
1.1.2 Internal Procurement Committees of a School	2
1.1.3 Global perspective on public procurement	3
1.1.4 Local perspective on public procurement	5
1.2 Statement of the problem	7
1.3 Purpose of the study	8
1.4 Research Questions	9
1.4.1 Main research question	9
1.4.2 Subsidiary research questions	9
1.5 Significance of the study	9
1.6 Definition of operational terms	10

1.7 Organisation of the study	11
1.8 Chapter summary	11
CHAPTER 2	13
REVIEW OF RELATED LITERATURE	13
2.0 Chapter overview	13
2.1 Management of public funds in secondary schools	13
2.2 Public procurement systems across the world	14
2.2.1 Public procurement regulations in secondary schools	14
2.3 Procurement methods	17
2.3.1 Open tendering	18
2.4.1.1 Open tendering principles	18
2.3.2 Request for proposal	19
2.3.3 Restricted tendering	19
2.3.4 Two stage tendering	19
2.3.5 Request for quotations	20
2.3.6 Single sourcing	20
2.4 The procurement process	20
2.4.1 Procedures for procurement of instructional materials in schools	21
2.5 Challenges facing the internal procurement committees	22
2.6 Measures that are deployed to tackle the challenges	24
2.7 Perception of internal procurement committees towards the procurement pro	ocess
	25

2. 8 Theoretical framework	26
2.9 Chapter summary	28
CHAPTER 3	30
RESEARCH DESIGN AND METHODOLOGY	30
3.0 Chapter overview	30
3.1 Research paradigm	30
3.2 Research design	31
3.3 Study area	32
3.4 Target population	32
3.5 Sample size	33
3.6 Sampling techniques	33
3.7 Data generation methods	34
3.7.1 Semi-structured interviews	34
3.7.2 Focus group discussions (FGDs)	35
3.7.3 Document Analysis	36
3.8 Data analysis procedures	36
3.9 Research Ethics	37
3.10 Limitations of the study	37
3.11 Chapter Summary	38
CHAPTER 4	38
RESULTS AND DISCUSSIONS	38

4.0 Chapter overview)
4.1 The public procurement policy in schools	9
4.1.1 Composition of the Internal Procurement Committees in schools4	0
4.2 Procedures that were being followed by the internal procurement committee	S
when purchasing goods and services	1
4.2.1 Procedures that are followed using open tendering method4.	3
4.2.2 Procedures that are followed when purchasing using request for quotations	5
4	4
4.2.3 Availability of written records of procurement in schools4	7
4.2.4 Benefits of implementing the public procurement policy in schools49	9
4.2.4.1 Transparency and accountability50	0
4.2.4.2 Purchase of goods and services of standard quality5	1
4.3 Challenges faced when implementing the public procurement policy in school	l.
52	
4.3.1 Unregistered suppliers52	2
4.3.2 Lack of finances53	3
4.3.3 Lack of training54	4
4.3.4 Cumbersome procedures in procurement5	5
4.4 Measures that were deployed to tackle the challenges	5
4.5 Perception of the procurement process by the internal procurement committee	e
59	
4.6 Chapter summary64	4

CHAPTER 5	65
CONCLUSIONS AND IMPLICATIONS	65
5.0 Chapter overview	65
5.1 Conclusion	65
5.2 Implications	66
5.3 Suggestions for further research	67
REFERENCES	69
APPENDICES	76

LIST OF APPENDICES

Appendix A: Authorisation letters to generate data	.69
Appendix B: Interview guide for head teachers	.70
Appendix C: Interview guide for IPC members	71
Appendix D: Focus group discussions guide for IPC members	72

LIST OF ABBREVIATIONS AND ACRONYMS

BSS Boarding Secondary School

CEED Central East Education Division

CGS Central Government Stores

CPAR Country Procurement Assessment Review

DSS Day Secondary School

GDP Gross Development Product

ICT International Competitive Tendering

IPC Internal Procurement Committee

JICA Japan International Cooperation Agency

MoEST Ministry of Education Science and Technology

NCT National Competitive Tendering

ODPP Office of Director of Public Procurement

ORT Other Recurrent Transactions

PPA Public Procurement Act

PPP Public Procurement Policy

PPR Public Procurement Regulations

TTC Teachers Training College

CHAPTER 1

INTRODUCTION

1.0 Chapter overview

This chapter introduces the study on the assessment of public procurement policy (PPP) implementation in secondary schools. It begins with a brief background of the procurement policy with a description of the development of internal procurement committees in public secondary schools. This is followed by the statement of the problem, purpose of the study and the research questions. It ends with the significance of the study and definition of some key terms.

1.1 Background to the study

In business management literature, public procurement is perceived as a major function of the government. Government performs major procurement activities in various sectors including education. Government entities, public procurement professionals and policy makers have paid much attention to procurement improvement. However, public procurement has been neglected in the academic and education research (Thai, 2001). It is against this background that the researcher intended to assess the PPP implementation in secondary schools.

1.1.1 The public procurement policy

There are several definitions of policy but for the purpose of this study, a policy is defined as an explicit or implicit single decision or group of decisions which may set out directives for guiding future decisions, or guide implementation of previous decisions (Haddad, 1995). Similarly, several authors have defined public policy in various ways but for the purpose of this study, only two definitions have been considered to act as a guide. Dye (2013) states that public policy "is whatever governments choose to do or not to do". Cochron and Malone (1995) similarly state that public policy "consists of political decisions for implementing programmes to achieve social goals". On the other hand, procurement is the activity of assessing, buying and receiving goods and services and it becomes public whenever this process is performed by public organisations or funded by public organisations. EL-Gayed (2013) defined the PPP as a statement or set of guiding rules, procedures or principles covering how procurement should be done within an organisation. From these definitions, PPP is a set of decisions, rules or procedures that guide public organisations in the purchase of goods and services. One of the organisations which use public funds to purchase goods and services is the school. The decisions regarding expenditure of public funds are made by the internal procurement committees (Mudolo, 2015).

1.1.2 Internal Procurement Committees of a School

Education institutions as public procuring entities are required by the PPR (2003) to establish IPCs to be responsible for procurement at various levels. Muchiri (2014), in his study on public procurement and disposal act and the performance of procurement in Kenyan Schools established that procurement committees are responsible for

procurement activities. The committee is constituted as follows; the deputy head teacher is the chairperson, the bursar is the deputy chairperson and at least six heads of department including the matron. The role of the internal procurement committees in schools is to ensure that there is prudence when handling school finances. As such, the IPCs are supposed to work in collaboration with other procurement units and the head teachers in accordance with the Public Procurement Act (2003). The key functions of the IPCs are to review, verify and ascertain that all procurement activities are in line with the procurement regulations (Mapulanga, 2015).

1.1.3 Global perspective on public procurement

In recent years, several countries have initiated processes of procurement of goods by following proper procedures with an aim of improving expenditure allocation on budget and management decisions thereby promoting good governance in government institutions (Stemele, 2009). Public bodies have always been big purchasers, dealing with huge budgets such that public procurement represents 18.42% of the world gross development product (GDP) (Migosi, Umbuki& Evusa, 2013). In this regard, most developed countries are engaged in public procurement reforms with the aim of increasing efficiency and improving service delivery although they do not usually require external assistance for the purpose of procurement reforms (World Bank, 2005). For instance, in the United Kingdom, public procurement has undergone various reforms aimed at enhancing efficiency and sustainable utilisation of public resources (Kiprop, Michael, Bornett & Lelan, 2015).

In developing countries, public procurement reforms were initiated by the World Bank following the need to fight corruption in the public sector, and as a condition for the provision of development aid (World Bank, 2016). Public procurement reform is being recognised as essential in service delivery such that it accounts for a high proportion of total expenditure allocated in the national budget. For instance, public procurement accounts for 70% in Uganda, 60% in Kenya and 58% of public spending in Angola. These figures are very high as compared to the global average of 12-20 % because of their prominent role in active infrastructure developments and social programs (Migosi et al., 2013).

The Government of Uganda initiated reforms in the public procurement and disposal sector in 1997, following the enactment of the 1995 constitution and the introduction of several reforms and structural adjustment programmes. The reforms were initiated against the backdrop of lack of accountability and transparency and absence of a culture of value for money in procurement activities. The development partners also recommended to Government to put in place appropriate public procurement practices based on international standards which are fair, transparent, competitive and non-discriminatory to all potential providers of goods, services and works.

Ghana underwent extensive public sector financial reforms in 1996 with the aim of improving public financial management so that public funds could be used effectively. In Kenya, the World Bank and the Kenyan government conducted a Country Procurement Assessment Review (CPAR) in 1997 and discovered many shortcomings in the Kenyan procurement system. The shortcomings included the absence of a sound regulatory framework, lack of competition and widespread abuses and fraud in the system. The CPAR found that the approach to procurement regulation limited the efficacy of public financial management which affected the government's ability to

deliver public services and there was no enforcement mechanism to enforce the rules which existed. The CPAR was thus the beginning of widespread reform in Kenya's public procurement system and the act was developed in 2005 (Benedict, 2017).

1.1.4 Local perspective on public procurement

Malawi is one of the developing countries that have also initiated the procurement reforms as a response to donors whose considered view is that adhering to approved and appropriate procurement procedures ensures that funds are used effectively to promote development (World Bank, 2005). Recent studies conducted by the World Bank have shown that almost 70% of development and recurrent expenditure in Malawi is realised through public procurement, hence it is important to regulate, monitor, and oversee public procurement in order to improve transparency, efficiency and value for money (World Bank, 2016). Actually, public procurement in Malawi is guided by the Public Procurement Act (PPA, 2003) and is defined as "the acquisition, by any means, of goods, works or services".

Procurement reforms in Malawi started in 2004 after the introduction of the new legal framework document, the Public Procurement Act (2003) which led to the decentralisation of the procurement process to the level of the public entity (World Bank, 2016). This implies that any public institution which is operational should always strive to spend on high quality goods and services at the lowest cost possible and maximise the use of goods or services acquired for the benefit of the institution. Basically, the reforms were done at institutional level to ensure that the general public that the institutions serve should benefit and be satisfied with the functions of the institutions (Majanga, 2015).

In 2007, the Office of the Director of Public Procurement (ODPP) designed strategies to sensitise employees in the education training and learning institutions on procurement procedures considering that these institutions get direct funding from the Ministry of Finance (World Bank, 2013). For instance, some secondary schools and approved community day secondary schools which are cost centres get funding from the treasury whose purpose is for Other Recurrent Transaction (ORT) in institutions (JICA, 2012).

Ideally, sensitisation meetings were necessary because public institutions like secondary schools collect public funds in form of school fees such that the funds are used for the running of the schools except the tuition fees which is deposited into the government account (Trigu, 2004). Thus, it was recommended that head teachers, who have the overall responsibility of managing school finances, should establish the Internal Procurement Committee (IPC) which should be responsible for the purchase of goods and services. Although the committees were established in schools, O'Neil, Cammack, Kanyongolo, Mkandawire, Mwalyambwire, Welham & Wild (2014) revealed that there were different procedures of procurement that were being followed in the education institutions. In this regard, sensitisation meetings were also conducted in 2015 such that head teachers, deputy head teachers and accounts clerks from all the schools in the divisions were trained on how to purchase goods and services following the procurement rules that require bidding and advertising to ensure effective use of financial resources (World Bank, 2016). In the Central East Education Division (CEED), the meetings were conducted at Kasungu Teachers' Training College (TTC) where head teachers were encouraged to ensure that all goods purchased at an

institution should be supported by proper documents to ensure effective use of public funds. As such, the trained personnel were given a task to train the IPCs in their respective schools to implement the PPP.

Regardless of the sensitisation meetings, public procurement practices in institutions have been characterised by huge contractual payment arrears, corruption and delay in delivery of services. In addition, audit reports showed that these institutions bypassed some procurement procedures to the extent of acquiring goods and services from non-existent suppliers at an expensive price as compared to the normal market prices. Moreover, most expenditure which was done at institutional level, especially on procurement of goods and services was not approved or not supported by proper documents (Majanga, 2015). In view of this, it was necessary to find out how public secondary schools were implementing the PPP.

1.2 Statement of the problem

There is an increasing consensus among researchers in the field of business management in both developed and developing countries that PPP implementation is absolutely crucial for the success of schools in financial management (Muchiri 2014, Doherty, McConnell and Ellis-Chadwick, 2013, Patrick, 2011). These studies have indicated that, in the developed world, this recognition has led to huge financial commitments in major initiatives to streamline the procurement function. However, little has been done in the developing world about the implementation of this policy in public institutions.

Malawi is one of the developing countries in Africa. The ODDP has been making efforts for several years to improve its procurement system in the education sector (Mapulanga, 2015). Sensitisation meetings were done in 2007 to improve the procurement systems in the schools. Mudolo (2015), in his study on the role of head teachers on financial management in Mzuzu City, stipulates that the IPC is supposed to allocate and distribute funds to sub programmes and activities of the school according to the budget. However, the study revealed that the procedures that the schools were following were different and contrary to the guidelines of school financial management. In addition, O'Neill et al (2014) also revealed that the procedures that the education institutions were following were different. This revelation prompted the ODPP to sensitise head teachers, deputy head teachers and accounts clerks or teacher bursars on the need to implement the PPP in schools. After the training that was done in 2015, there is lack of documentation to show how schools have implemented the PPP. It is in view of this that the researcher intended to assess how schools have implemented the PPP.

1.3 Purpose of the study

The purpose of the study was to assess the implementation of the public procurement policy. Specifically, the study intended to:

- 1. Evaluate the procedures which were followed by the internal procurement committees when purchasing goods and services.
- 2. Identify the challenges that the procurement committees faced when purchasing goods and services.
- 3. Establish measures that were deployed to tackle the challenges.

- 4. Determine the perception of the procurement process by the Internal Procurement Committee.
- 5. Establish reasons for holding these perceptions.

1.4 Research Questions

This research sought to answer the following main and subsidiary research questions.

1.4.1 Main research question

How do schools implement the public procurement policy?

1.4.2 Subsidiary research questions

The study was guided by the following questions;

- 1. What are the procedures that the internal procurement committees follow when purchasing goods and services in public secondary schools?
- 2. What are the challenges that the procurement committees face when purchasing goods and services following procurement procedures?
- 3. What measures are deployed to tackle the challenges?
- 4. How does the internal procurement committee perceive the procurement process?
- 5. Why do internal procurement committee members hold these perceptions?

1.5 Significance of the study

It is expected that the findings of this study would add to the stock of knowledge on procurement practices in schools since there is little published work on procurement in public schools in Malawi. It is hoped that the findings from this study would also inform policy makers and other stakeholders to identify areas that need their interventions with regard to designing changes to streamline the acquisition of goods and services in educational institutions as set by the PPA of 2003. Furthermore, the findings of this study would be beneficial to the head teachers and the IPC members to identify areas that can be improved for effective delivery of services in schools. Lastly, the study would also be useful to the supervisors of education management on how schools are managing their resources when it comes to purchasing of goods and services thereby ensuring prudence in the management of public funds.

1.6 Definition of operational terms

Corruption: This refers to influencing the decision making process of public officer or authority or breach of trust by a public officer in the exercise of his duty through fraudulent means like bribery and blackmail (Onyinkwa, 2013).

Cumbersome: This refers to something which is complicated because of being large, it is difficult to do or manage and it takes a lot of time and effort (Sarfo&Baah-Minta, 2013).

Fraud: This is a crime of getting money by misrepresentation of facts in order to influence a procurement process or the execution of a contract, and includes collusive practices among bidders, whether prior to or after bid submission, designed to establish bid prices at artificial, non-competitive levels and to deprive the procuring entity of the benefits of free and open competition (PPA, 2003).

Procurement: This is the process of acquiring goods, works and services, covering

both acquisitions from third parties and from in-house providers (Onyinkwa,2013).

Public funds: This means any monetary resources appropriated to procuring entities through budgetary processes, aid grants and credits put at the disposal of procuring entities by foreign donors, or revenues of procuring entities (PPA, 2003).

1.7 Organisation of the study

This study is organised into five chapters. The first chapter outlines the background of the study in terms of the general statement of the problem, purpose of the study, research questions guiding the study, significance of the study and definitions of operational terms. Chapter two presents the review of the related literature and the theoretical framework on which the study was based. Chapter three explains the research paradigm that was used in this study, sampling site and methods employed, how data was generated and analysed. Chapter four presents and discusses the research findings in line with the guiding questions, and finally chapter five presents the research conclusion and implications.

1.8 Chapter summary

This chapter has introduced the study on the assessment of PPP implementation in secondary schools. It has provided the brief background of PPP with a description of

the development of internal procurement committees in public secondary schools. In addition, it has explained the statement of the problem, purpose of the study, research questions and significance of the study. Finally, it has provided definitions of some terms that have been used in this study. The next chapter reviews the related literature.

CHAPTER 2

REVIEW OF RELATED LITERATURE

2.0 Chapter overview

This chapter reviews related literature on secondary school management, public procurement worldwide and in Malawi. In addition, it reviews the challenges that IPCs face when procuring goods and services and also the possible measures that can be taken to minimise the challenges. It also includes the perceptions of the IPCs towards the procurement process. Finally, it proposes the framework that locates the study within a body of literature on procurement.

2.1 Management of public funds in secondary schools

The secondary school management handbook in Malawi stipulates that head teachers have an overall responsibility of managing the school finances. To ensure proper use of public funds, it is the responsibility of head teachers to spend according to the budget, keep proper financial records and account for expenditure, auditing, and authorising expenditure (Government of Malawi, 2015). However, all these functions are not done by the head teachers alone; there are accounts clerks who are responsible for the collection of fees. In addition, the head teachers also set up an internal procurement committee which is responsible for allocation of funds to other sub-programmes at school level so that goods should be purchased following procurement procedures.

2.2 Public procurement systems across the world

The history of public procurement system dates back from between 2400 and 2800 BC as written on the red clay tablet found in Syria. In 800 BC, there was a silk trade between China and a Greek colony. In United States of America, goods and services that were needed by the government were being supplied by commissioners or commissaries. These suppliers received a commission on what they bought for administrative units. In late 1800s, the state legislature began to create boards and bureaus responsible for the purchasing of goods and services. During that time, central purchasing was a difficult practice but it became common in state and local governments. However, the system was challenged by some practitioners who contended that government should decentralise it so that support could be provided to end users (Thai, 2001). In addition, decentralization was preferred to eliminate bureaucratic obstacles and to empower managers to procure goods without problems from the central government. For this reason, there was a movement to adopt a uniform commercial procurement. This was done in 1951 and the commercial code was updated in 2000. Consequently, developed and developing countries have instituted their procurement regulations but the major obstacle is regulatory compliance.

2.2.1 Public procurement regulations in secondary schools

The PPP in schools in Kenya came into existence after the introduction of the public procurement disposal act in 2001. This was done due to mismanagement of funds by

school management boards and poor quality of work by some contractors (Muchiri, 2014). In 2002, the Ministry of Education embarked on measures to entrench provisions of the procurement regulations in academic institutions in 2002 to improve efficiency of procurement practices; thus, to enable schools manage expenditure and utilise public resources effectively. In this regard, circulars were sent to all public secondary schools, directing them to follow the new regulations to improve procurement practices and procedures. The new measures included establishment of school tender committees, training members of such committees, as well as principals, deputy principals, and staff directly involved in procurement activities. In the same year, initial training workshops targeting principals, deputy principals of national and provincial schools were organised by the Directorate of Public Procurement, with support from the Kenyan government (Kavula, Jeremiah, Kalai & Migosi 2014).

In the education sector, the procurement regulations and the Procurement Act are guided by the Public Procurement Manual for Schools and Colleges. The guide provides the primary legislative and policy frameworks for reforming procurement practices and managing procurement expenditure in public secondary schools. Actually, the regulations require public institutions to use standard tender documents, operate within set thresholds, and ensure that technical specifications meet international standards and that all bidders are treated equally irrespective of race, religion or nationality (Wambui, 2013; Ohene-Addae, 2012).

In addition, procurement regulations include the need for all tenders to be advertised widely in the print media, professional qualifications of bidders, effective record keeping, transparency in opening tenders, tender evaluation reporting, confidentiality of tender evaluation processes, as well as procurement planning and regulation of

emergency procurement. Furthermore, the regulations require school tender to have a membership of at least six heads of department or members of the teaching staff, including the Matron or officer-in-charge of the boarding facilities appointed by the school principal (Kotoka, 2012).

Even though the Government of Kenya has provided necessary legislative and policy frameworks to guide procurement reforms in all public institutions, empirical studies have shown than implementation of the PPP in public secondary schools across the country still remains a challenge. For instance, a study commissioned by the MOE of Kenya in 2006 revealed that more than half of secondary schools did not adhere to provisions of legislative and policy frameworks in their tendering processes. As a result, there was rampant corruption particularly at the administration and board levels with regard to procurement of school equipment, learning materials, supplies, and hiring of both teaching and non-teaching materials.

In Uganda, secondary school procurement is guided by the new public procurement and disposal asset guidelines of 2014. The aim of the guidelines is to create uniformity in the management of procurement and consistency using the best practices in order to achieve the value for money. Despite the guidelines, compliance to procurement policy still remains a challenge in public schools (Komakech and Machyo, 2015).

In Malawi, public secondary schools fall in the category of public entities as their procurement is funded by public funds (World Bank, 2005). In an effort to ensure that public funds are utilized effectively, Malawi has its procurement systems governed by the PPA of 2003 and public financial management act of 2003. However, there is no specific procurement manual for schools like in Kenya and Uganda. The role of

supervising and regulating public procurement in the country is carried out by the ODPP. However, depending on the institution, there are other assisting agents to ensure effective supervision. Thus for schools, the MoEST has its own supervisors who while performing their duties ensure that the procurement undertaken by the schools are done according to laid guidelines.

Public procurement regulations in secondary schools in Malawi were emphasized by the ODPP in 2007 considering that public schools use public funds to purchase goods and services (World Bank, 2013). However, O'Neil et al (2014), in their study on fragmented governance and local service delivery in Malawi revealed that education institutions were using different procedures when procuring goods and services. In view of this, sensitisation meetings were also conducted in 2015. The meetings targeted head teachers, deputy head teachers and accounts clerks or teacher bursars. This is different from what happened in Kenya whereby all individuals involved in procurement issues were trained. Although the procurement personnel were trained in Kenya and guidelines were issued in secondary schools, implementation of the PPP in schools has been criticised as inept, ineffective and corrupt and there are a number of factors that could lead to this characterisation. Considering that each country has different policy depending on its economic, social and political environment (Thai, 2001), the researcher's contention was to assess the implementation of PPP in Malawi as there was dearth of data on procurement in secondary schools.

2.3 Procurement methods

The preferred procurement methods are open tendering and request for proposal. However, there are other procurement methods which are also used. These are restricted tendering, two stage tendering, and request for quotations and single sourcing (Kotoka, 2012; Phoebe, 2008; PPR, 2004; ODPP Desk Instruction no. 3 of 2003). Different procurement methods are used under different circumstances so that the IPC approval is required before use of any method of procurement.

2.3.1 Open tendering

This is a default method of procurement for goods, works, and routine services. It is a competitive tendering method for goods, works and consultancy services of high value (Kotoka, 2012). The two types of competitive tendering methods are International Competitive Tendering (ICT) and National Competitive Tendering (NCT). ICT is used when effective competition cannot be achieved without the inclusion of foreign firms, and when the estimated value of the procurement exceeds the threshold value set by the ODPP. NCT is used when the procurement entity decides to do so and is subject to contract value thresholds.

2.4.1.1 Open tendering principles

An invitation to tender is published in the Gazette, local press and/or other media of wide international and national circulation. It is also displayed in the head office of the procuring entity in a publicly accessible place. Bidding documents are provided to all bidders responding to the invitation to tender or those that have been pre-qualified and the minimum bidding periods are 45 and 30 days for International Competitive Bidding and National Competitive Bidding respectively (Mapulanga, 2015)

2.3.2 Request for proposal

This is the default method of procurement for consultancy services. Normally, it starts with an expression of interest. Shortlisted bidders are at least three but no more than six. The proposals written include the minimum quality rating such that evaluation of the bidders considers both technical and financial status (Kotoka, 2012).

2.3.3 Restricted tendering

This is a procurement method with limited competition. A maximum of six and a minimum of three short-listed suppliers are invited to tender. To apply this method, a procurement entity requires specific approval from the Public Procurement Authority. This method is applied mainly when there is enough evidence to show that potential suppliers are limited, and that the procurement requirement is specialised in nature. In addition, restricted tendering is used when time and cost of considering a large number of tenders is disproportionate to the value of the procurement. The procedures for open tendering also apply to restrict tendering but the minimum bidding period is 21 days (Mapulanga, 2015; Kiprop et al, 2015).

2.3.4 Two stage tendering

Two-stage tendering is only allowed where detailed specifications cannot be made available before going to tender or the optimal solution is unknown. For instance, procurement of a hydro-electric plant. In the case of services, it is used to obtain proposals or offers on various means to meet its procurement requirements (Kotoka, 2012).

2.3.5 Request for quotations

This is the process of organising a fully-fledged tendering or other competitive tendering procedures. It is normally a time consuming and costly exercise which can be dispensed with when the contract value is small. The main idea is to allow procuring entities to use a simplified procedure in those instances where the contract value is small. The method is based on comparing price quotations obtained from at least three foreign or local suppliers for goods or works that are small in value. Bidders are given at least five days to respond to the request for quotation document. Thus, the procuring entity orders the goods and services by considering the lowest priced responsive bid (Muchiri, 2014).

2.3.6 Single sourcing

This system involves procurement from a supplier without any competition supplied by one source. It is normally used for procurement of sole or single sourced requirements. The law allows this method where procurement is for justifiably urgent items. A procuring entity negotiates with the supplier when using this method on quality, quantity, terms, price and time of delivery (Phoebe, 2008).

2.4 The procurement process

An individual procurement process differs slightly depending on the method of procurement and the type of procurement, that is, goods, works and routine services (Kotoka, 2012). However, the procurement process in public schools goes through five main steps which may vary according to different perspectives of looking at procurement. Basically, the first step is identification of the need so that preparation of

the specification by the IPC members can be done in line with the budget. This is followed by the preparation of the purchase requisition. The purchase requisition is checked and authorized by the relevant authority. Quotations are then obtained from which a supplier is selected, and then the issuance of purchase orders, thereafter goods are delivered or received. The final stage involves the receipt and checking of the invoice after which the payment is authorised and effected (Kimani, 2013; Onyinkwa, 2013).

2.4.1 Procedures for procurement of instructional materials in schools

To ensure effective use of public funds, the government of Malawi recommends that institutions should follow the procedures for procurement stipulated in the guidelines for the preparation for the 2015-2016 budgets when purchasing instructional materials. Prior to conducting procurement of goods, it is the responsibility of the procuring entity to ascertain whether the required supplies are available at the Central Government Stores (CGS) or in other government depots. When the CGS does not have the relevant items, it should request the concerned institution to get three quotations for the items from private suppliers through the usual IPC process. The CGS reviews the quotations and provides a "No Objection" waiver which enables the institution to undertake the procurement. This process can take a minimum of a day, depending on whether the quotations are provided in time (Malawi Government, 2015).

Although, the procurement procedures are outlined, studies have revealed that there are different procedures that are being followed in schools when purchasing goods and services (O'Neil et al., 2014; Michael, 2014; Migosi et al., 2013; Kimani, 2013; Onsongo, Okioga, Otiene and Mongare, 2012). O'Neil et al (2014) in their study on

fragmented governance and local service delivery in Malawi found that some of the local items like chalk were purchased at primary school level while some pens were being purchased at central system thereby duplicating the functions which is wasteful to the government. However, the study did not explain on what happens in secondary schools. Basically, several studies that were done in Kenya established that there was no uniformity in the procurement system for the public sector as a whole because schools were not familiar with the new procurement rules and regulations (Michael, 2014, Migosi et al., 2013; Kimani, 2013; Onsongo et al., 2012).

2.5 Challenges facing the internal procurement committees

Most developing countries are facing a problem of rapid changes in procurements which are imparting pressure on how the procurement function performs its internal and external processes in order to achieve its objectives (Wambui, 2013). Implementation of the PPP is driven by development partners who have focussed on getting countries to introduce a standard law developed by the United Nations Commission on International Trade Law (World Bank, 2005). The public procurement reforms currently focus on getting countries to implement a standard procurement law and system. This is a challenge that countries ought to cope up with as pointed out by Mulwa, Kalai & Migosi (2014) that public practitioners may have to put up with contradictory and contrasting procurement requirements and objectives imposed by policies and trade agreements. Such challenges in procurement reforms include processes, methods, procurement organizational structures and workforce which are beyond procurement regulations.

Public procurement in schools in developing countries faces many challenges. Several studies conducted in schools in Kenya, Uganda, South Africa and Ghana have indicated lack of training of head teachers and the IPC members, lack of support from the administration, long and cumbersome procedures of procurement, lack of funds, lack of regular auditing, and conflicts arising from the individuals as the challenges faced by the schools in the implementation of the policy (Rhamadan & Kibet, 2015, Kavula et al, 2014; Kusi, Aggrey & Nyarku, 2014; Michael, 2014; Onyinkwa, 2013; Kimani, 2013; Migosi et al., 2013; Nyakundi, Kombo, Omari & Mongare, 2012; Onsongo, et al., 2012). In addition, a study conducted by Rhamadan & Kibet (2015), on factors affecting implementation of public procurement procedures and practices in Kenya revealed that there were unethical behavioural practices done by some public officials and their friends which led to wastage of public resources and procurement of substandard goods that it did not serve the citizens appropriately.

Kusi et al (2014), in their study on assessment of PPP implementation in the education sector in Ghana established that there was a long procurement documentation process and unnecessary interference by heads of institutions which resulted in delays in the delivery of services. Similarly, Onyinkwa (2013), in her study on factors influencing compliance to procurement regulations in secondary schools in Kenya averred that some teachers who were trained as IPC members simply act as rubber stamps because some decisions concerning procurement were being made by head teachers without consulting the responsible members of the committee.

A study conducted by Nyakundi et al (2012) on challenges facing procurement committees in implementing the procurement act in secondary schools in Kenya

revealed that the procedures were still blanketed by confidentiality, disorganisation and dishonesty in various schools. This revelation is similar to the study done by Mapulanga (2015) on public procurement legislation and the acquisition of library materials in academic libraries in Malawi, who found that independent procurement lacked honesty so much that goods were procured at an expensive price. However, challenges in public schools in Malawi were not documented in his study as it only targeted libraries in higher academic institutions.

2.6 Measures that are deployed to tackle the challenges.

The secondary school committees that are responsible for procurement try their best to make the system more efficient regardless of the challenges mentioned. Several studies indicated that schools preferred to priotise when purchasing goods and services where funds were inadequate. If funds were not available, head teachers preferred to borrow from the suppliers and this resulted in single sourcing such that goods were procured at an expensive price (Kavula et al 2014; Michael, 2014; Mulwa, et al., 2014; Onsongo et al., 2012).

A study conducted by Mulwa et al (2014) on determinants of implementation of public procurement regulations in secondary schools in Kenya revealed that the level of compliance with the regulations was very low. They established that important factors like public procurement structure, induction of key personnel, trainings to principals and procurement committees need to be considered for effective implementation of PPP.

2.7 Perception of internal procurement committees towards the procurement process

Public procurement has been perceived as an area of waste and corruption in both developed and developing countries (Thai, 2001). In agreement, Basheka (2011) contends that public procurement is an area that is most vulnerable to corruption. Thus it is essential to establish a procurement system with clearly goals and objectives. Thai (2001), in his study on public procurement found that it was difficult to find statutes of procurement in small government entities where procurement budgets are small. The absence of a procurement code, regulations, or manual in such institutions leads to unpredictable procurement problems. For this reason, it is likely that one can hold unfavourable perceptions towards the procurement process.

Komakech and Machyo (2015), in their study on public procurement reforms in secondary schools in Uganda established that IPC members perceived the procurement process as an area for head teachers to use to find their personal needs rather than the needs of the institution. This was attributed to the fact that head teachers were receiving money from suppliers before the items were delivered in schools. Upon delivery of the items, quality was being compromised such that it was difficult for the IPCs to reject the delivered goods considering the friendship that existed between the supplier and the head teacher. For this reason, the IPC members lacked interest to monitor the procurement activities.

Onsongo et al (2012), in their study on challenges facing procurement committees in implementing the procurement act in Kenya, found that the procurement process was seen to be crucial because of the efficiency expected of school heads in the management

of school finances. This is the case because public procurement is a very complicated system within which there are many conflicting interests. Such being the case, implementers proposed that there should be sound procurement regulations in order to increase public confidence in the procedures followed in public procurement, and to ensure fair and equitable treatment of all.

2. 8 Theoretical framework

This study was guided by the institutional theory which was developed by Scott (Scott, 2004). Institutions are defined as social structures that have attained a high degree of resilience. The institutional theory is a theory which attends to the deeper and more resilient aspects of social structure. It considers the processes by which structures, including rules, norms, schemas, and routines, become established as authoritative guidelines for social behaviour. It inquires into how these elements are created, diffused, adopted, and adapted over space and time; and how they fall into decline and disuse. Institutions are made up of diverse elements that posit different bases of order, compliance and alternative rationales for establishing legitimacy claims. They are composed of regulative, normative and cultural cognitive elements that, together with associated activities and resources give stability and meaning to social life.

The regulative pillar emphasises the use of rules, laws and sanctions as enforcement mechanisms, with expedience as a basis for compliance. It generally uses some type of coercive force in order to gain compliance (Scott, 2008). The normative pillar refers to norms (how things should be done) and values (the preferred or desirable), social obligation being the basis of compliance. It defines what is right and appropriate. The cultural-cognitive pillar rests on the shared understanding with reference to what is

taken for granted. Scott explained that the three pillars may take prominence over time and circumstance although they are different.

Institutional theory explains how organisations are influenced by their environmental context. Scott (2004) contends that organisations of the same type tend to adopt strikingly similar organisational designs and behaviours as they share the same stakeholders and are subjected to the same pressure. However, the theory is generally static in nature and measures that are more direct are not developed in it.

This theory is a useful lens to assess the implementation of the PPP as it suggest that achieving legitimacy requires schools to conform to the government expectations that takes primacy over achieving efficiency. Dye (2013) explained that the government lends legitimacy to policies such that they are regarded as legal obligations which government institutions are supposed to abide by. Public schools are part of government institutions and are supposed to follow public policies.

In applying the institutional theory in this study, the school is an institution which is supposed to conform to the rules or procedures when purchasing goods and services in accordance with the procurement regulations. Government requires prudence in financial management in schools. It enforced schools to follow the procurement procedures when purchasing goods and services. Regulative compliance occurs when schools adapt to the regulations. In contrast, cultural cognitive compliance may arise if the policy is not clear and resources are not available such that decisions made depend on what the IPCs consider as a best practice without considering what the government wants them to do. Normative elements can be observed as a result of increased

knowledge and skills in procurement and proper communication of the policy in school. In this case, implementation of the PPP may be effective when the pillars in which schools rest are strengthened. This can happen if the procedures that are being used are properly enforced and developed. In addition, if the beliefs and values of head teachers and IPCs members provide more benefits and proper guidance (normative) and when the knowledge that could inform decision making for the IPCs to have positive perceptions towards implementation of the PPP is adequate and sufficient (cultural cognitive).

2.9 Chapter summary

This chapter has reviewed the management of public funds in secondary schools, procurement systems across the world and in secondary schools. It has discussed the methods that are used when purchasing goods and services. In addition, it has reviewed the procedures that are followed when purchasing goods and services and the challenges facing the internal procurement committees. The measures that were deployed to tackle the challenges and the perceptions of internal procurement committees towards the procurement process have also been reviewed.

Most of the studies that explored these issues were conducted in public schools in other countries but little is known on how public schools are implementing the PPP in Malawi. Moreover, the review showed that most studies used quantitative approaches which are in themselves not adequate in as far as the need to have an in-depth understanding of a phenomenon is concerned. In view of this, there was a need to conduct the study using the qualitative approach to have a better understanding of PPP

implementation in public schools in Malawi. The design and methodology of this study will be presented in the next chapter.

CHAPTER 3

RESEARCH DESIGN AND METHODOLOGY

3.0 Chapter overview

This chapter describes and justifies the research design and methodology that was employed to conduct the study. It also gives a description and justification of sampling, data generation tools, analysis and ethical considerations of the study. It concludes with the limitations of the study.

3.1 Research paradigm

A research paradigm is a cluster of beliefs that dictates the influence of what should be studied, how the research should be done and how the results should be interpreted (Bryman, 2008). The study was based on the interpretive paradigm such that it was qualitative. Interpretive paradigm was chosen for this particular study for two reasons. Firstly, it allowed the researchers to access the experiences and viewpoints of the research participants. Secondly, it recognised the role of the researcher and the research participants in knowledge construction, acknowledging interpretations as socially constructed realities. Using this paradigm, both the researcher and the respondents acquired active roles in the knowledge construction.

3.2 Research design

A research design is crucial in thinking how to guide data collection, its analysis and interpretation; how to establish a link between research question, data, analysis and conclusion; and how the research question can be addressed. For instance, Yin (2003) explained that a research design is a logical sequence that connects the empirical data to the study's initial research questions and conclusions.

The researcher used a case study design because it has several advantages. Firstly, it is able to capture and explore the complexity of phenomenon for a better understanding. This is what a large scale study like survey fails to achieve since it gathers superficial information about a phenomenon for the purpose of generalisation. Secondly, using the case study approach requires the use of multiple methods to collect data, enabling it to be validated through triangulation (Denscombe, 2003; Yin, 2003). Thirdly, it is action oriented and therefore the findings are useful for improving practice.

Considering the advantages, a case study design was used in order to develop in-depth analysis of the procurement procedures used by the IPCs in schools. Although a case study design is not a perfect design because it deals with processes that take place and their interrelationship (Kothari, 2004), it was preferred because it uses a small number of carefully selected samples to provide accurate and rich information about the topic at hand through in-depth examination of an issue. Creswell (2013) explained that a case study research is a qualitative approach in which the researcher explores a single case or multiple cases within a specified period of study. Though the objectives could be achieved through in-depth understanding of a single case, the researcher used multiple

case studies for the purpose of getting a broader understanding of the implementation of procurement procedures in schools.

3.3 Study area

The study was conducted in four secondary schools representing four cases which are recommended in qualitative research (Bryman, 2008). These schools were chosen because they had some common characteristics within the district. Firstly, they were very close to the division such that they had access to information regarding any changes that were necessary in schools. Secondly, they were close to the government stores where procurement of most school items was supposed to be done. In addition, these schools were also selected based on their enrolment capacity because enrolment of students determines the funds that could be collected and be used to purchase goods and services.

Kasungu District was selected because it was where the sensitisation meetings were conducted such that most representatives from the schools within the district managed to attend the meetings, thus the respondents were aware of the PPP. The area was also chosen because it was where the researcher resided, as such costs were minimised.

3.4 Target population

The target population is defined as the entire set of relevant units of analysis of data (Punch, 2005). The target population for the study was four public secondary schools. The respondents comprised all head teachers, deputy head teachers, accounts clerks or bursars and teachers who were members of the IPC.

3.5 Sample size

A sample is a portion or part of the population of interest (Kothari, 2004). The purpose of sampling is to gain an understanding about some features or attributes of the whole population based on the characteristics of the sample. Usually, the IPC of schools consist of a minimum of 3 members and a maximum of 7 members. The researcher sampled 6 respondents from each school making a total of 24. The sample consisted of 4 head teachers because they were controlling officers in all the activities that were carried out in the schools; 4 deputy head teachers who were the chairpersons of the IPCs; 2 accounts clerks from two schools and 2 teacher-bursars who were responsible for the collection of funds, and 3 teachers from each school who were IPC members because they were the ones tasked with the responsibility of overseeing the procurement process in their institutions.

3.6 Sampling techniques

Purposive sampling technique was used to select the respondents to permit the detailed study of part rather than the whole population based on experience of the respondents and objectives of the study. In addition, purposive sampling was used to select the major schools that were close to the town and had the larger enrolment. Creswell (2003) argued that in qualitative research, it is important for the inquirer to purposefully select individuals and sites that can provide necessary information. In this case, the researcher only targeted the respondents who were involved in the procurement process at school level.

3.7 Data generation methods

The researcher used several data generation methods. According to Patton (2002), using more than one data generation method strengthens and gives credibility to the study. The data generation methods that were used in this study were semi-structured interviews, focus group discussions and document analysis. Creswell (2013) elaborated that these methods are suitable in a qualitative study because they provide in-depth information to the research questions. These methods were used in order to have a deeper understanding of how schools were implementing the PPP.

3.7.1 Semi-structured interviews

According to Bryman (2008), a semi-structured interview refers to a process in which the interviewer has a series of questions that are in the general form of an interview guide but in which the interviewer is able to vary the sequence as well as ask further questions in response to what are seen as significant replies. Semi-structured interviews were preferred to provide the researcher with the main questions of focus which guided her to achieve the purpose of the study.

In addition, semi structured interviews were preferred to allow the researcher to ask probing questions where clarification was needed thereby addressing richness, depth of responses, comprehensiveness and honesty that are some of the hallmarks of successful interviewing.

Actually, the researcher interviewed 24 respondents. The respondents were 4 Head teachers, 4 deputy head teachers, 2 accounts clerks, 2 teacher-bursars and other 3 IPC members from each school. The focus was on procedures, challenges of the PPP, the

measures that were put in place to tackle the challenges and the perceptions of the internal procurement committees towards the procurement process. Two sets of openended questions were prepared in advance for the head teachers and IPC members. However, some questions in one set overlapped with those in another set. Follow up questions were also being asked for clarifications and elaborations. After asking for permission to interview the respondents, the researcher also asked if their views could be recorded. The interviewees accepted and their views were tape recorded. Field notes were also taken. The field notes described the setting in which the interviews were taking place and the non verbal communication for in depth understanding of the responses that were provided by the respondents.

3.7.2 Focus group discussions (FGDs)

Focus group discussions were done in all schools and the discussions targeted all the IPC members that were already interviewed. Actually, each group comprised 5 members who were IPC members. Head teachers were not part of the group because they were not IPC members. Open-ended questions were used to guide the researcher during the discussion. However, there was flexibility for follow up questions depending on the responses from the group. Focus group discussions were used to permit the researcher to collect data and insights that could be less accessible without the interaction of the group (Punch, 2005). Group discussions are important because they stimulate people in making explicit their views, perceptions and reasons thereby uncovering aspects of understanding that remain hidden. The FGDs helped to capture data on the perceptions of the IPC members and reasons for their perceptions. The results from the interviews and the FGDs were triangulated to ensure accuracy of the research findings (Creswell, 2013).

3.7.3 Document Analysis

Documents refer to those sources of data which are kept and are produced at the request of the researcher. The documents include such forms as letters, diaries, minutes, photographs, newspapers, magazines, videos, audios and autobiographies (Bryman, 2008). Such documents are important in case studies because they help to corroborate and augment evidence from other sources. In light of this, documents were reviewed to represent data to which participants give much attention although documents may not all the time be authentic and accurate (Creswell, 2003). The policy document was not available in schools. For this reason, the researcher only analysed the IPC minutes that helped in the interpretation of procedures used in procurement in schools.

3.8 Data analysis procedures

Data analysis is the process of bringing order, structure, and meaning to the mass of collected data. Through data analysis, the researcher addressed the research questions depending on the evidence collected. The researcher used thematic analysis to analyse the data as outlined by Braun and Clarke (2006). This procedure has seven stages namely: organising the data, familiarising yourself with the data, generating initial codes, searching for themes, reviewing themes, defining and naming themes and finally writing the report. Actually, the researcher read through all interview notes and listened to the interview records to acquire in-depth understanding of the data. The researcher categorised the collected data into meaningful categories developed from the interview questions. Then the researcher reduced and rearranged the data into manageable and comprehensive forms. After completing the process, the researcher continued to explore key themes and patterns or relationships among the data units. Thematic

analysis allowed the researcher to identify the themes that emerged from the responses of the 24 participants in schools. All the themes were linked to the research objectives.

3.9 Research Ethics

In conducting a qualitative study, Creswell (2013) advises researchers to seek and obtain permission from the authorities in charge of the site of the study because it involves a prolonged and extensive data collection. Initially, a letter of introduction from Chancellor College was obtained. The researcher used the introductory letter to seek permission to collect data from the schools from the Education Division Manager in CEED. The researcher then took the letter from the Division and the college to the head teachers of the concerned schools so that participants should be prepared for what was ahead. The head teachers welcomed the researcher and informed the IPC members about the purpose of my visit. The IPC members read the letters and accepted to participate in the study. Despite the fact that the participants read the letters in advance, verbal consent was sought from all the participants during the data collection for them to be flexible in responding to the questions. When analysing the data, names of the specific schools from which the data was collected were withheld for confidentiality purposes. The schools were coded as BSS1, BSS2, DSS1 and DSS2.

3.10 Limitations of the study

Yin (2003) notes that every research study is limited by the constraints placed upon the researcher and this research is no exception. The study was limited to four schools in

Kasungu District. The results must be treated with caution as they cannot be generalized to all public secondary schools in Malawi. This was done due to limited time and financial resources. However, the results could be used as a pointer for a more comprehensive study on PPP implementation in schools in Malawi because government schools follow the same regulations. The issues featured in 4 schools are likely to be repeated in other schools.

3.11 Chapter Summary

This chapter has provided the in-depth account of the research design and methodology employed in this study. The study was based on interpretive paradigm and the case study design was chosen to realise the research objectives and questions. The data was generated using face-to-face semi-structured interviews, focus group discussions, and documents were also reviewed. Thematic analysis was chosen as a means to analyse the data. The researcher sought the consent of all participants involved in the study. Finally, limitations of the study have also been explained in the chapter. The next chapter presents and discusses the research findings according to the objectives of the study.

CHAPTER 4

RESULTS AND DISCUSSIONS

4.0 Chapter overview

This chapter presents the data collected and their analyses. Themes that emerged during the analysis of data from the interviews, focus group discussions and document review were all linked to the research questions. The study answered five research questions. Hence, the presentation of the findings will follow the five questions showing how the data collected answered the questions. However, the researcher will first provide an understanding of the PPP by the participants.

4.1 The public procurement policy in schools

When respondents were asked to explain their understanding of the PPP in schools, their responses were not different in all the schools. One of the interviewees explained that:

"PPP is a set of procedures which should be followed when schools would like to purchase goods and services using public funds" (Head teacher BSS1).

Another head teacher defined the policy in this way:

"These are guidelines or procedures that are followed when purchasing items at institutional level" (Head teacher, BSS 2).

One IPC member explained that:

"PPP are guiding rules and principles on how schools should purchase their items from vendors" (IPC member, DSS 1).

Another IPC member defined the policy as follows:

"This is a policy that guides IPCs on how to purchase goods and services at an institution using public funds" (IPC member, DSS 2).

From the definitions above, it clearly shows that the interviewees were aware of the PPP. In general, the PPP was defined as a set of procedures which are used to purchase goods and services in institutions using public funds. The policy requires that each institution should have the IPC that should be responsible for the purchase of goods and services. This understanding of the policy is in line with El-Gayed (2013) who argued that it is the statement or set of guiding rules, principles or procedures on how procurement should be done within the organisation. Considering that policies are regarded as legal obligations which schools as public institutions are supposed to comply with (Dye, 2013), it was found that each school that was under study had the IPCs which were responsible for the purchase of goods and services. This agrees with the findings by Muchiri (2014) who established that schools' IPCs are to guide in the purchase of goods and services.

4.1.1 Composition of the Internal Procurement Committees in schools

All the schools had IPCs as per the requirement of the PPA (2003).

The members of the committees comprised:

- The Deputy Head teacher as the chairperson
- The accounts clerk/ teacher- bursar
- Heads of department

This implies that IPC members were automatically chosen based on the responsibilities assigned to them. According to the institutional theory, the schools applied the

normative pillar which relies more on internalisation processes. As such, it was less likely for the IPCs to induce only lip services. However, at BSS 2, some of the heads of department were not members of the IPC. Instead, some teachers were included as IPC members. The head teacher of this school revealed that the IPC was set by the former head teacher. He explained that:

I am new in the office as the head teacher although I have been serving here for seven years as the deputy. I am not aware of why the former head teacher chose to exclude some of the heads of department as IPC members. I think he based on their dedication to duties. I am aware that heads of department are supposed to be members of the IPC (Head teacher, BSS 2).

The statement that was made by the head teacher at BSS 2 indicated that this school valued the cultural cognitive pillar which rests on a more deeply set of beliefs and assumptions to ensure that this policy was being implemented. This implies that the former head teacher used his common beliefs on how teachers were committed to their work without considering what the PPA (2003) stipulated regarding the composition of the IPCs.

4.2 Procedures that were being followed by the internal procurement committees when purchasing goods and services

The first question sought to assess the procedures that were being followed by the internal procurement committees when purchasing goods and services. During the interviews, it was revealed that the procedures that were being followed in schools depended on the procurement method that was used. This agrees with Kotoka (2012)

who established that individual procurement slightly depended on the method of procurement and the type of goods and services being procured.

At BSS1 and BSS2, the open tendering method and request for quotation method were being used to purchase their goods and services. The researcher then asked why they preferred to use these methods. One IPC member explained that:

"The school purchases a lot of foodstuffs such that it is necessary to carefully compare the prices and the quality of food from different suppliers" (IPC Member, BSS1).

This implies that the methods were preferred because they provided room for comparison of prices such that goods were purchased from the lowest evaluated supplier. One interviewee highlighted that:

I may tell you that we follow open tendering and request for quotation methods to purchase goods and services but normally some teachers are the main suppliers at this school such that it is difficult to determine what is followed on the ground (IPC member, BSS2).

The researcher further asked the respondents why teachers were supplying goods at the school instead of using open tendering method. In his response, he explained that:

Open tendering method is perceived as bureaucratic and inflexible because the prices that are offered by the suppliers are too high to cater for the fluctuations in prices of the products. This method requires buyers and suppliers to work in partnership to ensure that public funds are being used accordingly and we still have a long way to go for the school to meet the requirements (IPC member, BSS2).

At DSS 1 and DSS 2, the commonly used method was request for quotation but sometimes the tendering method was also used at DSS 1. The request for quotation

method was preferred because it provided the actual prices of goods and services at that particular period unlike the tendering method. However, single sourcing sometimes was also used depending on the need at the school although it was not recommended due to its lack of competition from different suppliers. All the respondents revealed that the revenue that was being collected per term was not enough for schools to comply with the request for quotation method every time they wanted to purchase goods and services. This means that the request for quotation method was partly being used such that some items were being purchased using single sourcing. This statement is in line with Ohene- Addae (2012) who found that most schools resort to single sourcing instead of competitive bidding. This implies that the policy was sometimes ignored due to unavailability of funds. As such, goods were being purchased on credit from one supplier.

4.2.1 Procedures that are followed using open tendering method

Before schools acquired and used any resource, they considered the importance of the goods and services to be procured. During the interviews, one respondent revealed that:

"The items are lined up according to the priorities by the department in need and the head teacher approves the document so that payment vouchers can be written by the accounts clerk" (IPC member, BSS1).

Another IPC member narrated:

"At the beginning of the term, each department comes up with the items that are needed for that particular term. The items are then scrutinized by the IPC members during the IPC meeting. The items considered are then submitted to the head teacher for approval" (IPC member, BSS2).

From the explanations above, it is clear that BSS1 and BSS2 prioritise the items needed before they purchase their goods and services. 12 respondents revealed that the first thing that IPCs considered most was advertising where by the IPC chairperson put up posters on the notice board and on trees surrounding the schools so that suppliers can apply. There is a special day which is set for opening the box containing the applications from different tenders. The respondents indicated that they announce the tenders in the presence of the tenderers or their representatives so that there should be fairness in the evaluation. Then the IPC members meet and evaluate the bidders depending on the quality and price that they have written in the tender. After selecting the bidders, phone calls are made to inform them about their success. Then letters are sent to notify them that they have qualified to be suppliers for the school within a particular period. Upon delivery of the goods, payment is done using closed cheques to ensure that accountability is adhered to.

At BSS1, it was revealed that this process was normally done in August when the students were on holiday so that suppliers should be aware before the beginning of the school academic calendar which is September. At BSS2, there was no specified period that this process was being done. The IPC members revealed that they used open tendering method when there was a need. If there was no bidder at that particular time, they used request for quotations or single sourcing to purchase food stuffs.

4.2.2 Procedures that are followed when purchasing using request for quotations

10 respondents at BSS1 and BSS2 explained that they commonly used request for quotation method when purchasing teaching and learning materials if they found that the items they wanted were not available at the government stores. On the other hand, 11 respondents at DSS1 and DSS 2 revealed that the IPC considered the need for the specified items. The three quotations were sought from vendors in town from which selection was based on the quality and prices indicated. The IPC then wrote the purchase order. When delivery of goods was done, cheques were then issued by the head teachers to the supplier. Sometimes, teachers were sent to buy the items if the suppliers could not manage to bring their items to the school. However, one IPC member at DSS1 revealed that:

The three quotations method is used but sometimes all these quotations are obtained from one supplier who is registered and has the knowledge of the public procurement policy. To ensure that they should not lose customers, some suppliers who are not even registered have three different quotation books. They write the three quotations with slightly different prices so that goods should still be bought from their shop (IPC member, DSS1).

Another IPC member explained that:

"We usually go to town and purchase most of the items from one shop. Some items are purchased from local vendors but we liaise with the shop owners to include them on the quotation" (IPC member, DSS2).

From the explanations above, it implies that there was corruption when purchasing goods and services using request for quotations method. As a result, goods might be purchased at a more expensive price than the acceptable price which was paid by the general public. This was contrary to Onyinkwa (2013) who emphasised that goods should be procured in a way that allowed for fair and open competition. In addition, suppliers might arrange with the person responsible for the purchase of items to inflate

prices so that the buyer and the supplier should also benefit from the transactions. This is corruption despite the government's commitment to reduce corruption.

At DSS2, respondents indicated that the three quotations requirement was not adhered to sometimes because it was time consuming and most of the items that were bought using this method were expensive. The head teacher at DSS2 gave this as an example:

"When we wanted to construct a classroom block, the quotations that we received from three contractors charged ten times more than the local builders. Considering that the school had meagre resources, we chose not to use this method. Instead, local builders were hired to do the work and the classroom block was constructed" (Head teacher, DSS2).

This implies that the request for quotation method was not being commonly used because suppliers were offering their goods at an expensive price as compared to the normal market price. This is in agreement with Muchiri (2014) who indicated that the request for quotation method was a costly exercise especially when the purchase was of small value. In accordance with Scott (2004), the findings have shown that BSS1, BSS2 and DSS1 were institutionalising themselves to standardize the procedures that were reinforced by the government but at DSS1, regulatory compliance was not easy due to poor resources and expensive prices that suppliers were offering. As a result, they were procuring their goods based on what they took as the best method in their own understanding. Hence, it can be argued that procurement was not being carried out as required due to the prices that suppliers were offering on the market and available resources therefore compromising the implementation of the policy.

4.2.3 Availability of written records of procurement in schools

During the interviews, the researcher sought to find out if the schools had written records for procurement transactions at the school. In their responses, all the interviewees at BSS1, BSS2, DSS1 and DSS2 indicated that they kept records for all procurement transactions that were being done in schools. The researcher then reviewed the documents that were in the procurement files in all the schools. During the review, it was noted that the IPC minutes were available in the files in all schools. The minutes showed what the IPCs discuss before purchasing goods and services. In addition, supporting documents like quotations, delivery notes, invoices were available for some items purchased at BSS1, BSS2 and DSS1. At DSS2, only a few quotations were available because they were purchasing most of the items locally.

When respondents were asked about the policy document on procurement, they revealed that there was no written document from MoEST concerning the public procurement policy in all schools. 12 respondents revealed that schools did not have the copy of the policy as it was verbally communicated during the sensitisation meeting such that the IPC members were just briefed by the head teachers. However, 3 head teachers from BSS1, BSS2 and DSS1 revealed that the documents on how schools were supposed to purchase goods and services were provided during the meeting at Kasungu Teachers Training College but they were not printed. One head teacher explained that:

I was trained at the TTC for three days. On the last day, the presenter informed us that only soft copies were available and it required one to have a flash disk. By then, I did not have that flash disk to get the information. After the training, I did not take an initiative to get the documents from my

colleagues or from the division because I was not fully conversant with procurement issues though I went for the training (Head teacher DSS2).

This showed that there were some institutional requirements which were strongly backed by head teachers while some received sympathetic responses to implement them. On the other hand, 2 head teachers from BSS2 and DSS1 revealed that most of the IPC members did not show interest to read the soft copies that were available because they had a lot of work to do. Similarly, 4 IPC members from BSS2 and DSS1 elaborated that they did not show interest in reading the documents. This prompted the researcher to find out why they did not have that interest. One IPC member from DSS2 averred that:

"I am the IPC member but I act as a figure head in the committee. Most of the items that are purchased at this school are done by the head teacher, deputy head teacher and the accounts clerk" (IPC member, DSS2).

Another IPC member explained that:

I am not actively involved in purchasing goods especially food stuffs. Most of the items are purchased from some teachers here. When I try to inquire more on how goods are being purchased, I am threatened by the head teacher and being told to wait for my time when I will be promoted to the headship position. For this reason, I remain quiet because it is the head teacher who is the controlling officer at the school (IPC member, BSS2).

This means that the IPCs were not really in a position of implementing the procurement policy due to their lack of involvement in some procurement issues. The threats they received from their head teachers were also limiting them from performing their duties effectively. This concurs with Onyinkwa (2013) who explained that some teachers who were trained as IPC members simply acted as rubber stamps because some decisions

were made by head teachers without consulting the responsible members. This implies that 3 schools had some tangible evidence and appropriate records of procurement. However, there was negligence on the part of head teachers to produce hard copies of the notes so that IPC members and teachers could easily access the information for transparency and accountability. These findings support the institutional theory by Scott (2004) who explained that institutions are influenced by the environments. The threats that the head teachers throw on the IPC members limited the members from full participation in procurement issues thereby compromising implementation of the PPP.

In addition, the IPC members were not doing their job accordingly as indicated by Mapulanga (2015). This was evidenced by the lack of interest of some IPC members to read and understand the documents that were presented by the ODPP officials during the training. This implies that some IPC members were not participating in the procurement process. They lacked strength to regulate decisions that go against institutional interest. As such, they were not in a position to fully implement the policy. This behaviour might create a leeway for manipulation of things by head teachers, deputy head teachers and accounts clerks. As such, massive losses of public funds could be incurred.

4.2.4 Benefits of implementing the public procurement policy in schools

The researcher sought responses as to what the schools benefited when implementing the PPP. The responses were transparency and accountability and purchase of goods and services of standard quality.

4.2.4.1 Transparency and accountability

Respondents at BSS1 and DSS1 argued that the policy promoted transparency and accountability such that other teachers who were not IPC members had a clear mind on what was happening at the school. However, 6 respondents from BSS2 and DSS2 revealed that the IPC's did not work for the intended purposes. Most of the expenditure was done by a few members and sometimes only the head teachers, deputy head teachers and bursars were responsible for the purchase of goods and services. One respondent revealed that:

The IPC members usually meet and scrutinize the items that are needed in different departments but when the goods or services are purchased, it is not known to all members if all the items planned for were available or not. Only a few members are aware of the transactions. In addition, the total expenditure of the goods and the receipts are not shown to some members but I know that the policy can promote transparency and accountability if it can be implemented properly (IPC member, DSS2).

This implies that the respondents were aware that effective implementation of the policy could promote transparency and accountability but they chose not to adhere to the requirements. Sometimes head teachers, deputy head teachers and accounts clerks used the regulative pillar of institutionalization which emphasized on the use of rules, laws or sanctions to implement the public procurement policy where expedience was the basis for compliance. However, this was not the intended purpose for the decentralisation of the procurement reforms which led to the establishment of the IPC. As such, there was a possibility of fraud in some schools. According to Sarfo and Ba-Mintah (2013), transparency has been defined as the creation of an environment where

information on existing conditions, decisions and actions are made visible and accessible to the individual participant. If this assertion is to go by, then procurement in schools was not really transparent since some members did not have the detailed procurement information. It implies that implementation of the policy was being taken for granted following the cultural cognitive pillar of institutionalisation to suit individual interests such that there might be misappropriation of public funds.

4.2.4.2 Purchase of goods and services of standard quality.

12 respondents from BSS1 and DSS1 revealed that the public procurement policy helped to purchase goods and services of good quality. This concurs with the study by Onsongo.et al. (2014) who affirmed that goods and services that were being procured in schools by following the procurement procedures were of standard quality. However, 12 respondents from BSS2 and DSS2 argued that quality was being compromised because of late payment of school fees by some students such that goods were bought locally and favouritism sometimes existed, hence low quality things were being purchased. Thus, it can be argued that implementation of public procurement policy could assist schools to purchase goods and services of standard quality if the procedures were followed properly and if students paid their school fees on time. In addition, implementation could be effective if head teachers and IPCs worked with integrity at all times.

4.3 Challenges faced when implementing the public procurement policy in school.

The role of PPP in schools is to ensure that there is regulatory compliance and prudent financial management. Institutional theory suggests that coercive governmental pressure would push the authorities in institutions to adopt the policy. Unfortunately, the PPP implementation faces various challenges in public secondary schools. The researcher sought to find out the challenges that the IPCs faced when implementing the PPP. Respondents pointed out that lack of registered suppliers, lack of finances, lack of training and cumbersome procedures in procurement were the major challenges in all the schools. The challenges have been discussed in the following sections.

4.3.1 Unregistered suppliers

All respondents indicated that most of the suppliers were not registered as recommended by the PPA (2003). As such, most of them were not willing to give out quotations because they were not familiar with procurement procedures. For this reason, 15 respondents revealed that they negotiated with one supplier who was registered to write three different quotations just to fulfil the procurement procedures. However, 5 respondents highlighted that the procurement procedures did not fit with the community that the schools served. One interviewee gave an example that:

It is difficult to purchase locally available materials like brooms following the procurement procedures because most of the vendors are from the community who do not really have the necessary documents like the licence and the ODPP certificate to support their business. Such being the case, the items are purchased from the vendors but payment of cheques is done through teachers who have got identity cards and bank accounts because the requirement is that institutions

should issue closed cheques. However, open cheques are also used (IPC member, DSS1).

This indicated that it was difficult to purchase items from vendors using the procurement procedures. As such, teachers were being involved to use their accounts to withdraw money for the vendors. This behaviour may create chances of making unfavourable decisions to manipulate figures so that teachers can also benefit from the transactions considering that teachers have their own work to do rather than withdrawing money on behalf of the vendors. This concurs with the study by Ohene-Addae (2012), who revealed that the PPP created more difficulties which eventually created the chances for corruption and fraud. This finding suggested the need for appropriate procurement methodologies that can even suit the vendors whose capital for business was minimal so that implementation of the policy in public schools could be beneficial to the development of the nation. This implies that an improvement in the public procurement process could in no doubt create wealth and reduce poverty in the country as a whole.

4.3.2 Lack of finances

Many respondents at BSS1, BSS2 and DSS1 revealed that the ORT that is received from the government is minimal as compared to the needs of the schools such that schools relied on local revenue that was collected as school fees. In addition, many respondents at BSS1, BSS2 and DSS1 explained that the resources from the local revenue were not enough to procure goods following the procurement procedures because most students were not paying their school fees on time. At DSS2, respondents

revealed that the school was a non-cost centre such that it only received grants from the government. One respondent at this school explained that:

"Procurement of some goods and services sometimes is done even before the IPC met to scrutinise the budget since purchasing was based on the priorities that the school need when the term begins" (IPC member, DSS2).

This means that expenditure of public funds at DSS2 was being done without considering the budget, which was contrary to MoEST (2014) who emphasised that expenditure of public funds should be done according to the budget. From an institutional point of view, the IPC members of this school seem to be entrenched in their views. In addition, all respondents in all schools explained that there were many delays in funding such that their planning was not being accomplished. Moreover, they argued that it was difficult to pay the suppliers accordingly if most students failed to pay school fees on time. This implies that only the basics of procurement were being done in some schools. Thus the procedures were not really being followed as recommended in the PPA (2003) as evidenced by delayed funding, late payment of school fees, and purchasing of things before the IPC met to scrutinise the budget.

4.3.3 Lack of training

Many IPC members indicated that it was very difficult to follow the procurement procedures as recommended because most of them lacked knowledge on procurement. Many of the respondents said that they were trained as teachers and their job was to teach. One respondent elaborated that:

I was trained as a teacher at college and nothing was said on procurement. My expectation was only to teach the learners and nothing else on procurement. By virtue of being the head of department I was told that I am a member of IPC. Without any training, I have been purchasing goods for the school for two years such that I usually use common sense instead of the procurement procedures. I do not know how long it takes for the government to train IPC (IPC member, DSS2).

This is in agreement with Onsongo et al. (2014) who revealed that the secondary school procurement committees lacked sufficient information on procurement procedures such that goods and services were purchased using different procedures depending on the knowledge that the committee had. This implies that lack of training was a challenge in schools where there were no procurement personnel who could guide the members in the purchase of goods and services. Training the IPC members was very important so that they might be transformed from being struggling amateurs to skilled members because they would be knowledgeable, confident and more active (Michael et. al. 2014). Considering the knowledge gap that existed among some IPC members and the head teachers, there was a possibility that the procurement procedures could easily be compromised.

4.3.4 Cumbersome procedures in procurement

Respondents at BSS1, BSS2, DSS1 and DSS2 revealed that procurement procedures were cumbersome and time consuming such that IPC members spent most of the time sourcing quotations instead of teaching. One respondent revealed that:

It is difficult to use the procurement procedures all the time because it is a long process and time consuming more especially when there is an emergency case like sickness and funeral (Deputy head teacher, DSS1).

Another respondent narrated:

The procurement process is too long for a teacher to follow it. It consumes most of the teaching time whenever the school tries to follow the proper procedures especially when a child is sick during the night or death of a member of staff (IPC member, BSS2).

This implies that there was some sort of bureaucracy in the procurement process which made it cumbersome such that it was difficult for one to be assisted during emergencies. This concurs with Michael et al. (2014) who revealed that procurement procedures were cumbersome and time consuming such that teachers were spending most of their time in purchasing goods and services instead of teaching. While the government had good intentions to save public funds through implementation of this policy, the findings clearly show that there are more challenges than expected. As a result, schools tend to use their own ways provided they purchase goods and services as required without considering the actual procedures that should have been followed per the requirement of the PPA of 2003.

This implies that implementation of the public procurement policy in schools was done based on the common beliefs that the institution had. This shows that schools were using the cultural-cognitive pillar to implement the PPP because of the long process involved in procurement.

4.4 Measures that were deployed to tackle the challenges

The researcher sought to find the measures that were deployed to tackle the challenges. The findings revealed that IPCs that were responsible for procurement tried their best to make the system more efficient regardless of the challenges discussed. Respondents at BSS1, BSS2 and DSS1 revealed that the challenge of unregistered suppliers is handled by negotiations. They revealed that they negotiated with the suppliers to issue three quotations as required even though they knew that the items were purchased using single sourcing method. In one school, respondents explained that:

"Suppliers seem not to be aware of this policy. We tried to explain to them on how schools should purchase goods from their shops and they assist us with the three quotations as required" (FGD centre, BSS1).

In another school, respondents confided:

From the explanation at BSS1, it is clear that schools try their best to consider the regulatory pillar of institutionalization by explaining to the suppliers about the need to have three quotations. On the other hand, the findings from DSS1 clearly showed that there was corruption and fraud for the two parties to reach a compromise. Respondents at DSS2 revealed that they purchased most of the items locally to avoid wastage of time with the suppliers in town.

The challenge of lack of finances was being tackled by considering the needs of the school. 18 respondents revealed that they preferred to prioritise when purchasing goods and services where funds were inadequate. If funds were not available, they preferred to borrow from the suppliers even though the goods were procured at an expensive price. This concurs with reviewed studies which established that head teachers preferred to borrow from suppliers. This resulted in single sourcing such that goods were procured at an expensive price (Kavula et al. 2014; Michael, 2014; Onsongo et al., 2012). With the meagre resources that the schools collected, many respondents at DSS2 revealed that they purchased most goods and services locally. During the focus group discussions, respondents at one school explained that:

There is a common practice of emergency procurement especially at the beginning of the term at this school when students fail to pay school fees on time but the goods and services at that particular time are needed most. In this case, the IPC addresses the situation at hand by purchasing goods and services locally (FGD Centre, DSS2).

From the explanations above, it implies that the revenue that was being collected in some schools was not wisely used through the available procurement procedures. The single sourcing that was being preferred to as a solution to tackle the challenge of finances can enable institutions to save public resources if properly used. However, this practice may lead to massive loss of public funds if not used properly, for example, through inflated figures. Procurement of goods as an emergency case as indicated at DSS2 can also lead to massive loss of public funds through misguided priorities.

On the issue of training as a challenge, head teachers, deputy head teachers, accounts clerks and teacher bursars revealed that they use the little knowledge they got during

the three days training they had in 2015 at Kasungu TTC. One head teacher explained that:

It is not easy to procure goods and services following the procurement policy because I am not an expert in procurement. Some of the procedures can be followed properly if schools had experts in procurement. In addition, I only attended training in 2015. There has been no training since that time (Head teacher, DSS2).

This statement clearly shows that it was difficult to effectively implement the policy because the head teacher who was supposed to train the IPC members was not much conversant with procurement issues. As such, it was likely that the IPC members too would have problems in understanding the procurement issues. This implies that the cultural cognitive pillar dominated most when handling procurement issues in schools unlike the normative pillar which focuses on what is right and appropriate.

Lastly, 18 respondents revealed that the procurement policy is open to misinterpretation depending on how the institution perceived it. In this regard, there was little consideration to implement this policy effectively because procurement is a long process. For this reason, 4 respondents revealed that they usually purchase goods locally and get receipts from vendors to support their expenditure although this is not recommended. Such being the case, there is a need to simplify the procedures and make them clear to all the IPCs in public schools to ensure proper implementation of the public procurement policy.

4.5 Perception of the procurement process by the internal procurement committee

During the focus group discussions, the general perception of 18 respondents was that the public procurement policy is a good policy especially in this era of democracy. It is one way of dealing with corruption in institutions but the way it was being implemented in schools was contrary to what the government expected. At one school, respondents elaborated that:

Procurement is a wide topic which requires more time for us teachers to understand the necessary procedures. The head teachers, deputy head teachers and accounts clerks were trained for three days and we understood that there was much content that was delivered on these days such that those who attended the training did not manage to capture the important information clearly. When they came here, they only briefed us for some hours and yet they had several documents that were given by the ODPP officials. We think there should be proper arrangements that the whole IPC of public schools and the head teachers should be trained accordingly (FGD centre, DSS1).

Another school explained that:

The PPP should be looked at further; public procurement is problematic in schools. Look at our school, it is close to the town but we are still failing to implement the policy. What about schools in rural areas? Can they manage to follow the procedures if schools in town are failing? Corruption will still continue if there will be no changes (FDG centre, BSS2).

Respondents from BSS1 and DSS2 echoed similar sentiments that there was a need to look at the procurement system further so that implementation of the PPP can achieve its intended purpose. This implies that the PPP implementation was difficult because the people who were given the tasks in schools were not fully conversant with procurement issues and the procedures required some amendments to fit in schools.

There was a need for proper training for all the members involved in procurement so that they could work with confidence as professionalism is one of the incentives that can promote implementation. According to Scott (2004), implementation of the policy is effective if the actors are well conversant with what they are doing. The revelation that the members who attended the meeting failed to capture all the important concepts showed that the training was done for a short period of time. In addition, the short briefing that was done at school level implies that those who attended the training were not fully conversant with the procurement issues.

4.6 Reasons for holding the perceptions towards the procurement process

The fifth objective was to establish the reason why the IPCs hold their perceptions towards the procurement process. When respondents were asked to give the reasons why they hold their perceptions towards the procurement process, most of them gave reasons that were related to poor communication of the policy, lack of training, lack of procurement manuals and inadequate resources in schools. During the focus group discussions, respondents at one school uttered that:

Respondents at BSS1 explained that:

Yes we know the importance of the policy especially in this era of democracy but we still need some more knowledge of procurement process. It's difficult to implement this policy because we do not even have the act that guides the procurement. We wonder why the government wants us to implement this policy and yet it is not serious to train us and give us the necessary documents................................ Others received money for the training they attended but as IPC members we did not have that chance (FGD centre, BSS1).

At DSS1, respondents narrated:

Policies need to be communicated to us in a proper channel before they want us to implement them. This PPP seems to be poorly implemented because we were not even involved to explain our concerns on what really happens in schools regarding school expenditure. All what came to us is that we need to follow proper procedures when procuring goods and services, some of us have never seen the PPA which can guide us on how we should purchase things. We think MoEST in conjunction with the ODPP should provide procurement manuals to schools to act as a guide. In addition, the government should increase funding and provide to us on time (FGD centre, DSS1).

At DSS2, respondents explained their reason in this way:

We hold negative perceptions towards the implementation of this PPP. We have never heard our head teacher explaining to us that there are sufficient funds available for a particular term. All we hear is that funds are always insufficient to cater for all school activities. We tend to wonder why the government wants us implement this policy and yet the funds are inadequate in schools. Moreover, the grant that we receive from the government is very minimal and most of the time there are some delays to access the funds. This

policy requires schools to have adequate funds for it to be implemented. Without this consideration, loopholes will remain in schools (FGD centre, DSS2).

From the explanations above, it was clear that the implementation of the PPP was negatively affected by lack of proper communication from MoEST to the schools, lack of proper training, lack of procurement manuals for the IPCs and inadequate finances in schools. Because of these problems, respondents felt that implementation of the PPP could not be effective in schools.

From an institutional stand point, it can be inferred from the data that head teachers considered the PPP as an important legal obligation such that IPCs were institutionalized in all schools. However, the procedure that schools were using to purchase their goods and services indicated that the pillars in which the schools rested were weak. Cultural cognitive element dominated in most of the decisions that were being used to conform to the PPP. Moreover, the perceptions that the IPCs held and the reasons they gave indicated that PPP was not fully institutionalised in schools. Furthermore, the PPP was poorly enforced by the government evidenced by lack of the circular in schools. For this reason, implementation of this policy was based on the cultural cognitive pillar because of unclear policy and poor enforcement from the government. The government placed higher coercive pressure for the institutions to undergo a process of implementing the policy but the rate of adoption in schools is generally very low. This clearly shows that sanctions cannot work better when implementation is based on what the implementers perceive as important in their own understanding.

4.6 Chapter summary

This chapter has presented and discussed the findings of the study in the light of the analysis of the data collected. The major findings presented in this study include the procedures that were being followed in schools when purchasing goods and services. The challenges that the internal procurement committees faced when implementing the policy have also been discussed. In addition, the study has presented and discussed the measures that were deployed to tackle the challenges in the perceptions of the internal procurement committees and the reasons why they hold their perceptions towards the procurement process.

CHAPTER 5

CONCLUSIONS AND IMPLICATIONS

5.0 Chapter overview

The study was set out to assess the public procurement policy implementation in selected secondary schools in Kasungu District. This study was undertaken purposely not only to fulfil the academic pursuit but also to ascertain the actual procedures that were being used in public schools when purchasing goods and services to ensure prudence in financial management so that all stakeholders in education could also be informed on the matter. This chapter presents the conclusion and the implications of the study. It also highlights some areas for further research that the current study did not address.

5.1 Conclusion

Based on the findings of the study, the researcher concluded that the public procurement policy was increasingly being recognised in schools as a policy that could play a significant role in the management of public resources in education institutions. However, the procurement procedures that were adopted in public schools were quite different and not actually in line with the procurement guidelines although public schools share same goals, regulatory environments, structures and needs. Implementation of the public procurement policy seemed to be challenging in public

schools because the ones handling the procurement activities were not fully trained and that public schools could not afford to follow all the procedures due to lack of finances. To tackle the challenges, schools were purchasing their goods and services using single sourcing such that goods were being procured at an expensive price. Moreover, there was cheating when sourcing out the three quotations because quotations were being obtained from one supplier upon agreement, instead of three different suppliers. This is also corruption. According to institutional theory, the pillars that the schools were resting on seem to be weak evidenced by a widespread use of the cultural cognitive pillar. Proper implementation of this policy could be one way of dealing with corruption in public schools thereby improving financial management of public funds. Therefore, it can be argued that the public procurement policy is partially being implemented in public secondary schools

5.2 Implications

If schools will continue using different procurement procedures when purchasing goods and services using public funds then there might be imprudent financial management. The implication is that it will be difficult to monitor and account for public funds thereby failing to achieve the national education standard of financial management.

The procurement procedures used in public schools should be reviewed and guidelines should be provided so that eminent difficulties facing proper implementation of this policy can be addressed. Basically, there is a need to specify how goods can be purchased from vendors whose capital for business is minimal so that implementation of the policy in public schools can be beneficial to the development of the nation.

Some IPC members in schools are not fully empowered to perform their duties effectively. This means there may be misappropriation of public funds by some head teachers, deputy head teachers and accounts clerks or teacher bursars who are considered as key players in public schools. This implies that IPCs should be empowered so that they should have greater control of procurement activities at school level. When the items are purchased, all the members should have access to information regarding the purchasing process so that there should be transparency and accountability in schools. In addition, IPC members and suppliers should work in partnership so that goods that are purchased should be of good quality regardless of the price fluctuations on the market. This can promote prudence in financial management.

A system to provide continuous training to head teachers and IPC members who are the ones to implement the policy is highly recommended. It is important that MoEST should develop a procurement training program for head teachers and IPCs of schools. All the IPC members need to be exposed to further training so that they should be knowledgeable, confident and more active. The training programmes are required to equip the IPC members with skills in order to achieve the procurement policy objectives. In addition, suppliers need training in order to increase transparency and reduce corruption. If some IPC members are not fully conversant with the procurement procedures then there may still be unprocedural practices when using public funds. For this reason it could be difficult to achieve the national goals of education in Malawi.

5.3 Suggestions for further research

The present study used a very small sample and the results cannot be generalised. It is necessary for future studies to explore the issues raised in this study on a broader scale to understand what the situation is like in Malawi. In this study, only schools that were close to the CEED offices and those in town were sampled. It is necessary to find out how schools in other divisions and those in rural areas are implementing the policy, and compare with those from the urban areas.

REFERENCES

- Basheka, B.C. (2011). Compliance to public procurement reforms in developing countries: the contextual perspective from Uganda. *International journal of procurement management (IJPM)* 4(5), 535-548.
- Benedict, K.K. (2017). Principal's characteristics influencing implementation of procurement practices in public secondary schools in Embakasi sub county, Nairobi county, Kenya (Master's thesis). University of Nairobi, Kenya.
- Braun, V. & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative* research in psychology, 3(2), 77-101.
- Bryman, A. (2008). Social research methods (3rd ed.). Oxford: Oxford University Press.
- Cochron, C., & Malone, E. (1995). *Public policy: Perspectives and choices*. New York: McGraw-Hill.
- Creswell, J. W. (2003). Research design: Qualitative, quantitative and mixed method approaches. California: Sage Publications Limited.
- Creswell, J. W. (2013). Research design: Qualitative, quantitative and mixed method approaches (4thed.). California: Sage Publications Limited
- Doherty, N.F., Mc Connell D.J., & Ellis-Chadwick, F (2013). Institutional response to electronic procurement in the public sector. *International journal of public sector management (IJPSM)*, 26(6), 495-515.
- Dye, T.R. (2013). *Understanding public policy*. (14th ed.). California: Prentice-Hall Inc.

- EL-Gayed, Y. (2013). The influencing factors of public procurement policy development: The case of Libya (Doctoral Thesis). United Kingdom, The University of Salford.
- Government of Malawi (2003). Desk instructions for public procurement. Lilongwe: Ministry of Finance.
- Government of Malawi (2004). *Public procurement regulations*. Lilongwe: Ministry of Finance.
- Government of Malawi (2015). *National education standards: Primary and secondary education*. Malawi: Ministry of Education
- Government of Malawi. (2003). *Public procurement act*. Lilongwe: Ministry of Finance.
- Haddad, W. (1995). *Educational policy planning process: An applied framework*. Paris: UNESCO.
- JICA, (2012). Basic education sector analysis. Malawi: International Development Centre of Japan Inc.
- Kavula, M.J., Jeremiah, M., Kalai, J.M., & Migosi, J. (2014). Determinants of implementation of public procurement regulations in Kenyan secondaryschools. *International Journal of Educational Research and Reviews* (IJERR), 2 (1), 009-016.
- Kimani, W. R. (2013). Factors influencing the procurement Practices in Public Secondary Schools in Mathioya. (Master's thesis). University of Nairobi, Kenya

- Kiprop, C., Michael, J.J., Bomett, E., &Lelan, J. (2015). Issues in the implementation of instructional materials procurement policy in Kenyan public primary schools:

 The case of Wareng District, Uasin-Gishu County. *International Journal of Advanced Research (IJAR)*, 3(10), 1806-1816.
- Komakech, R.A. & Machyo, J. (2015, August 24-28). *Public Procurement Reforms: A Disaster for Public Secondary Schools in Uganda:* Paper presented at the International Conference on Good Governance and Service Delivery in Developing Economies, held in Kampala, Uganda
- Kothari, C.R. (2004). Research methodology, methods and techniques. India: New Age International Ltd.
- Kotoka, F. (2012). Assessing the level of compliance with the public procurement act 2003(act 663) in public entities in Ashanti region (Master's thesis). Kwame Nkrumah University of science and Technology, Ghana.
- Kusi, L.Y., Aggrey, G.A.B., &Nyarku, K.M., (2014). Assessment of public procurement policy implementation in the educational sector: A case of Takoradi Polytechnic. *International Journal of Academic Research in Business and Social Sciences (IJARBSS)*, 4(10), 1226-1245.
- Majanga, B.B (2015). Sealing leakages of public funds abuse. The Malawi cash gate case. *Applied Finance and Accounting Journal (AFAJ)*,1(2), 726-735.
- Malawi Government. (2015). *Guidelines for the preparation of the 2015/2016 budget*.

 Malawi: Ministry of Finance, Economic Planning & Development.

- Mapulanga, P. (2015). Public procurement legislation and the acquisition of library materials in academic libraries in Malawi. *Library review*, 64(1/2), 101-117
- Michael, J.J., Stephen, K.R., &Kiprop, C. (2014). Challenges facing school board management in the implementation of the procurement policy in Kenya. *Global Journal of Interdisciplinary Social Sciences*(*GJISS*), 3(3), 104-112.
- Migosi, J.A., Umbuki, C. N., & Evusa, Z. (2013). Determinants of noncompliance of public procurement regulations in Kenyan schools. *International Journal of Education Administration and Policy Studies (IJEAPS)*,5(7),154-159.
- Muchiri, K.P. (2014). Public procurement and disposal act and the performance of the procurement function in public secondary schools in Nairobi county(Master's thesis). University of Nairobi, Kenya
- Mudolo, V. (2015). The Head teachers' Financial Management Role in Mzuzu City

 Public Secondary School (Master's thesis). University of Malawi, Chancellor

 College, Zomba
- Mulwa, J.K., Kalai, J.M. & Migosi, J. (2014). Determinants of implementation of public procurement regulations in Kenya's secondary school. *International Journal of Educational Research and Review (IJERR)*, 2(1), 009-016
- Nyakundi, E., Kombo, C., Omari, R., & Mongare, O. (2012). Challenges facing procurement committees in implementing the procurement act; a case study of secondary schools procurement committees in Kiisi County, Kenya. *Elixir International Journal (EIJ)*, 48(1), 9186-9191

- O'Neill, T., Cammack, D., Kanyongolo, E., Mkandawire, M.W., Mwalyambwire, T., Welham, B. A., & Wild, L. (2014). Fragmented governance and local service delivery in Malawi. London: Overseas Development Institute
- Ohene-Addae, A. A. (2012). An assessment of the impact of efficient procurement practices on the corporate performance of a public university: A case of Kwame Nkrumah University of Science and Technology (Master's thesis). Kwame Nkrumah University of Science and Technology, Ghana.
- Onsongo, E.N., Okioga, C.K., Otiene R.O., and Mongare, O. (2012). Challenges facing procurement committees in implementing the procurement act: A case study of Kisii County. *Elixir International Journal (EIJ)*, 48 (2012), 9186-9191
- Onyinkwa, J. (2013). Factors influencing compliance to procurement regulations in public secondary schools in Kenya. A case of Nyamache District, Kiisi County.

 Interdisciplinary Journal of Contemporary Research in Business (IJCRB), 5(1), 1-32
- Patton, M. Q. (2002). *Qualitative research and evaluation methods*(3rded). London: Sage Publications Limited.
- Phoebe, B. (2008). The public procurement system in South Africa: *Main Characteristics*. *Public Contract Law Journal (PCLJ)*, 37(4), 781-802.
- Rhamadan, K.& Kibet, Y.(2015). Factors affecting implementation of public procurement procedures and practices in Elgeyo-Marakwet county.

 International Academic journal of procurement and supply chain management (IAJPSCM)

- Sarfo, A.P., and Baah-Mintah, R. (2013). Assessing the effect of the procurement act (663) on the public financial management in Ashanti Region. *American Journal of Rural Development(AJRD)*,1(4), 91-98.
- Scott, W. R. (2004). *Institutional theory: Encyclopaedia theory*. California: Sage Publication Limited.
- Scott, W. R. (2008) *Institutions and organizations: Ideas and interest*. California: Sage Publication Limited
- Stemele, B.M. (2009). Assessing good governance in procurement at Lejweleputswa District Municipality. (Master's thesis). Stellenbosch University, South Africa.
- Thai, K.V. (2001). Public procurement re-examined. *Journal of public procurement*, 1(1), 9-50.
- Trigu, D.A.M.N (2004). The impact of the Malawi secondary school cluster system on the management of community day secondary schools: Case of Blantyre District (Master's thesis). University of Massachusetts, United states of America.
- Wambui, E. N. (2013). Role of procurement on organization performance; A survey study of public secondary schools in Imenti North District, Kenya. *International Journal of Social Science Entrepreneurship (IJSSE)*, 1(3),289-302.
- World Bank (2005). Harmonising donor practices for effective aid delivery: strengthening procurement capacities in developing countries. United States of America: OECD publishing.
- World Bank (2013). Procurement capacity assessment and strategy formulation in Malawi. A case study. United States of America: OECD publishing.

Yin, R.K. (2003). Case study research: Design and methods (3rd ed.). California: Sage Publications.

APPENDICES

AppendixA: Authorisation letters to generate data



CHANCELLOR COLLEGE

Principal: Richard Tambulasi, B.A. (Pub Admin), BPA (Hon), MPA, Ph.D

Our Ref.: EDF/6/19 Your Ref.:

27th February 2017

P. O. Box 280, Zomba, MALAWI Tel: (265) 01 524 222 Telex: 44742 CHANCOL MI Fnx: (265) 01 524 046

TO WHOM IT MAY CONCERN

INTRODUCTORY LETTER FOR MASTER OF EDUCATION (Policy Planning & Leadership)

Ms. Beatrice Nyongolo Mangochi (MED/PPL/26/15) is a student of Education in the Department of Education Foundations at Chancellor College, University of Malawi studying for her Masters in Policy Planning and Leadership.

She is working on her thesis titled, "Assessment of Public Procurement Policy Implementation in Secondary schools in Kasungu District".

This is meant to be a request to your institution or organization to assist our student in her endeavor to collect data.

Thank you

F. A MTEMANG'OMBE

audisaulou lo

HEAD-EDUCATION FOUNDATIONS

UNIVERSITY OF MALAYM CHANCELLOR COLLEGE DEPT OF EDU, FOUNDATION 2017 -02- 27 PO BCC 280 ZOMSA

University of Malawi Chancellor College P.O Box 280 Zomba 06th March, 2017

The Education Division Manager

Central East Education Division

Private Bag 233

Kasungu

GRANTED THUMPHU (PEMA) EDUCATION DIVISION
OF THE
CENTRAL EAST

2017 - U. U. 6

PRIVATE BAG 25
KASUNGU

Dear Sir,

PERMISSION TO COLLECT DATA IN SELECTED SCHOOLS IN KASUNGU DISTRICT.

I would like to request for A permission to collect data in the following schools:

- Kasungu Secondary School
- · Chayamba Secondary School
- · Chankhanga Community Day Secondary School
- · Chiphaso Community Day Secondary School

I am a postgraduate student pursuing a Master's degree in Policy Planning and Leadership at Chancellor College, University of Malawi. I am conducting a research for my final year project, which is a requirement of the degree programme titled "Assessment of Public Procurement Policy Implementation in Secondary Schools in Kasungu District as per attached introductory letter.

The information that will be collected will solely be used for academic purposes and will be treated with utmost confidentiality. Any assistance rendered on the basis of this information will be highly appreciated.

Yours faithfully,

Beatrice Nyongolo Mangochi

Appendix B: Interview guide for head teachers

- 1. What is your understanding of public procurement policy?
- 2. Do you have the internal procurement committee?
- 3. Who are the members of the IPC at this school?
- 4. How are the members of the IPC selected at this school?
- 5. How often do they meet for procurement proceedings?
- 6. Do you keep records for the procurement proceedings?
- 7. What are the guidelines which support the procurement process from the Ministry of Education?
- 8. What are the procurement methods that are used when purchasing goods and services?
- 9. What steps are followed at this school when procuring goods and services?
- 10. What do you consider when approving procurement plans for the school?
- 11. What benefits have accrued from the implementation of public procurement policy in your school?
- 12. What are the challenges faced in the implementation of the public procurement policy?
- 13. How do you tackle with the challenges to ensure effective use of public funds?
- 14. Do you think the policy has improved financial management in your school?
- 15. If you had a chance of advising the government about the policy regarding procurement in public schools, what would you say?
- 16. Do you have any issue/s you would like to add or explain that has not been covered in the questions?

Appendix C: Interview guide for IPC members

- 1. What is your understanding of public procurement policy?
- 2. Do you have the internal procurement committee?
- 3. How are the members of the IPC selected at this school?
- 4. What are the guidelines which support the procurement process from the Ministry of Education?
- 5. What are the procurement methods that are used when purchasing goods and services?
- 6. What steps are followed at this school when procuring goods and services?
- 7. What benefits have accrued from the implementation of public procurement policy in your school?
- 8. What are the challenges faced in the implementation of the public procurement policy?
- 9. How do you tackle with the challenges to ensure effective use of public funds?
- 10. What are your perceptions towards the procurement process?
- 11. Do you have any issue/s you would like to add or explain that has not been covered in these questions?

Appendix D: Focus group discussions guide for IPC members

- 1. How often does the procurement committee meet?
- 2. Do you always implement the public procurement policy when purchasing goods and services at this school?
- 3. Why do you hold some perceptions towards the procurement process?
- 4. What are the major issues that are of great concern regarding the procurement process at this school?
- 5. If you had a chance of advising the government about the policy regarding procurement in public schools, what would you say?
- 6. Do you have any other thing you want to say that I have not asked you?